Author's response to reviews

Title: Iodine Nutritional Status after the Implementation of the New Iodized Salt Concentration Standard in Zhejiang Province, China

Authors:

Yan Zou (zouyan0573@sina.com)
Xiaoming Lou (xmlou@cdc.zj.cn)
Gangqiang Ding (zouyan0573@gmail.com)
Zhe Mo (zhmo@cdc.zj.cn)
Wenming Zhu (difang901@yeah.net)
Guangming Mao (difang903@yeah.net)

Version: 7 Date: 6 August 2014

Author's response to reviews: see over
Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Iodine Nutritional Status after the Implementation of the New Iodized Salt Concentration Standard in Zhejiang Province, China (Manuscript Number: 2020010636125692). Those comments are all valuable and very helpful for revising and improving our paper. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked using the track changes mode. The main corrections in the paper and the responds to the reviewer’s comments are as flowing:

Responds to the reviewer’s comments:

Response to Editors:

We have invited a native English speaker reviews our manuscript.

Responds to the reviewer’s comments:

Abstract:

We have revised the abstract according to reviewer’s suggestion.

We are very sorry for our negligence in preparing the manuscript and we have made corrections on the manuscript. We appreciate for Editors/Reviewers’ warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for your
comments and suggestions.

Sincerely,

Yan Zou

Zouyan0573@sina.com